Author's response to reviews

Title: Using targeted subsidies to improve access to skilled birth attendance for poor women: a case study of vouchers plus health equity funds in three rural health districts in Cambodia

Authors:

Por Ir (irpor@yahoo.com)
Dirk Horemans (dirk.horemans@btcctb.org)
Narin Souk (pbhs.kc.soc@everyday.com.kh)
Wim Van Damme (wvdamme@itg.be)

Version: 2 Date: 12 July 2009

Author's response to reviews:

MS: 7089701028272958
Research article
Using targeted subsidies to improve access to skilled birth attendance for poor women: a case study of vouchers plus health equity funds in three rural health districts in Cambodia
Por Ir, Dirk Horemans, Narin Souk and Wim Van Damme
BMC Pregnancy and Childbirth

Dear Editor,

Following your request, we have made change to our manuscript by adding a paragraph to clarify the Ethical Approval for the study at the end of the method section on page 14 as follows:

Ethical consideration
This study is part of the EC-funded Poverty and Illness research project (www.PocIll.com), which received ethical approval from the national Cambodian Ethical Committee. Verbal informed consent was obtained from each participant and respondent before the focus group discussion and interview.

Best regards,

Authors